Author's response to reviews

Title: Development and validity of a questionnaire to test the knowledge of primary care personnel regarding nutrition in obese adolescents

Authors:

Lucinéia Pinho (lucineiapinho@hotmail.com)
Paulo Henrique T Moura (paulohtm@yahoo.com.br)
Marise F Silveira (ciaestatistica@yahoo.com.br)
Ana Cristina C Botelho (anacristina.botelho@yahoo.com.br)
Antônio P Caldeira (antonio.caldeira@unimontes.br)

Version: 4 Date: 1 May 2013

Author's response to reviews:

Dear editor

We answered to the questions raised by the Associate Editor and hope to have met his requests. The answers are listed below, indicating where the corrections were made.

As suggested by the editor, the text underwent another review by a language editing service and a native English speaker. Therefore, if writing still requires improvement, we would appreciate any recommendations for another language editing service or precise indication of text fragments that should undergo English polishing.

Given the abovementioned, we ask you to consider again the revised version of the manuscript for publication in BMC Public Health.

Yours sincerely,

Lucinéia de Pinho and co-authors